Site visit 11/18

Got the finest dusting of snow overnight, partly cloudy today.

After packing up and cleaning the cabin we paid a last visit to the site. Data check this morning showed no pwrmon data from cl starting yesterday around
noon, so we went to the spares room again and grabbed a spare victron and victron cable. Then out to site around 9. Started by swapping out the body of
Ethan's lidar pi 4 with the body of pi 8, which seems to have a better wifi signal inside the same box. Pi 8 doesn't have a cable to connect the GPS to, but
Ethan says that's not a problem. Then looked at victrons. We suspected that the victron usb cable in cl may just have been loose or unplugged, as we had
been climbing all over the lower part of C yesterday and could easily have nudged it. Will reseated both ends of both cables, and then (once | remembered
| had to stop and start nidas to get usb devices to show up) data was flowing again. So, as far as we know, everything is working at site as of 10:30 today.
Also while at site we added a bunch of zip ties to the barrel, and took some site pics.

Out to Mt. Crested Butte by 11:15, home to snowy Boulder by 4:15. Thanks to Steve and Gary for their remote help the past couple days!
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